

Geophysical Research Letters

RESEARCH LETTER

10.1029/2018GL078105

Kev Points:

- Chemical alteration of glaciated volcanic bedrock is dominated by silica dissolution and precipitation
- · Dissolved silica in glacial meltwater is greater at more mafic study sites and results in deposition of opaline silica rock coatings
- Past glacial chemical weathering may be responsible for some amorphous silica deposits on the highly mafic surface of Mars

Supporting Information:

- Supporting Information S1
- Data Set S1

Correspondence to:

A. M. Rutledge. arutled@purdue.edu

Citation:

Rutledge, A. M., Horgan, B. H. N., Havig, J. R., Rampe, E. B., Scudder, N. A., & Hamilton, T. L. (2018). Silica dissolution and precipitation in glaciated volcanic environments and implications for Mars. Geophysical Research Letters, 45. 7371-7381. https://doi.org/10.1029/ 2018GI 078105

Received 29 MAR 2018 Accepted 6 JUL 2018 Accepted article online 13 JUL 2018 Published online 1 AUG 2018

©2018. The Authors.

This is an open access article under the terms of the Creative Commons Attribution-NonCommercial-NoDerivs License, which permits use and distribution in any medium, provided the original work is properly cited, the use is non-commercial and no modifications or adaptations are made.

Silica Dissolution and Precipitation in Glaciated Volcanic **Environments and Implications for Mars**

A. M. Rutledge¹ (D), B. H. N. Horgan¹ (D), J. R. Havig² (D), E. B. Rampe³ (D), N. A. Scudder¹ (D), and T. L. Hamilton⁴ 问

¹Department of Earth, Atmospheric, and Planetary Sciences, Purdue University, West Lafayette, IN, USA, ²Department of Earth Sciences, University of Minnesota, Twin Cities, Minneapolis, MN, USA, ³NASA Johnson Space Center, Houston, TX, USA, ⁴Department of Plant and Microbial Biology, University of Minnesota, Twin Cities, Minneapolis, MN, USA

Abstract The surface of Mars exhibits strong evidence for a widespread and long-lived cryosphere. Observations of the surface have identified phases produced by water-rock interactions, but the contribution of glaciers to the observed alteration mineralogy is unclear. To characterize the chemical alteration expected on an icy early Mars, we collected water and rock samples from terrestrial glaciated volcanics. We related geochemical measurements of meltwater to the mineralogy and chemistry of proglacial rock coatings. In these terrains, water is dominated by dissolved silica relative to other dissolved cations, particularly at mafic sites. Rock coatings associated with glacial striations on mafic boulders include a silica-rich component, indicating that silica precipitation is occurring in the subglacial environment. We propose that glacial alteration of volcanic bedrock is dominated by a combination of high rates of silica dissolution and precipitation of opaline silica. On Mars, cryosphere-driven chemical weathering could be the origin of observed silica-enriched phases.

Plain Language Summary The planet Mars has glaciers and ice sheets on its surface and probably did in the past. Minerals on the planet's surface form in the presence of water, but it is unclear which minerals may have formed due to liquid water under warm climates versus those formed under much colder climates. In order to study this problem, we collected rocks and water from Mars-like analog sites: glaciated volcanoes. We measured the chemistry of the water and the mineralogy and chemistry of rock coatings found near the glaciers. Both the water and the rock coatings were high in silica. We propose that glaciers alter volcanic bedrock by dissolving and precipitating noncrystalline silica. Silica detected on the surface of Mars could have formed due to similar processes.

1. Introduction

Water ice has significantly modified the surface of Mars, based on features that have been interpreted as residual glacial landforms or ground ice deposits (Lucchitta, 1984; see also Colaprete & Jakosky, 1998; Head et al., 2005; Mangold, 2003; Milliken et al., 2003; Pierce & Crown, 2003; Rossbacher & Judson, 1981; Sharp, 1973; Squyres, 1978, 1979; Squyres & Carr, 1986). Geomorphic and radar data support the presence of near-surface, present-day ice, including mantled scarps with deposits of water ice >100-m-thick, glacier-like forms, viscous flow features, and lobate debris aprons, some of which are thought to be debris-covered glaciers (Dundas et al., 2018; Holt et al., 2008; Levy et al., 2007, 2010, 2014; Pierce & Crown, 2003; Plaut et al., 2009; Souness et al., 2012). These glaciers most likely formed during climate excursions during the Middle to Late Amazonian (Head et al., 2006; see also Fassett et al., 2014; Fastook & Head, 2014) and were covered with debris during or toward the end of their emplacement (Fastook et al., 2014). However, the areal extent and duration of glaciers and ice sheets on the surface of early Mars is poorly constrained.

Recent climate modeling favors a cold and icy scenario for early Mars (Forget et al., 2013; Kite et al., 2014; Wordsworth et al., 2013), leading some to propose an early Mars primarily affected by extensive glaciation with transient ice and snow melting (Cassanelli & Head, 2015; Fastook & Head, 2015; Kite et al., 2013; Wordsworth et al., 2013). However, it is unknown what chemical alteration a long-lived cryosphere and associated intermittent glacial meltwater would have caused on the mafic surface of Mars. Understanding the distribution of mineral deposits due to glacial weathering on Mars will help us understand past climates of Mars. The nature of the early Martian climate is a critical unresolved issue in planetary science.

<mark>-</mark>





Figure 1. Locations of field sites in the Cascades Range, northwestern United States. The yellow dots indicate water sample locations at each individual site, and the red dots indicate the rock and sediment samples of interest in this study. (a) Gotchen Glacier; (b) Eliot Glacier; (Ci) Collier Glacier; and (Cii) Diller and Hayden Glaciers.

On Earth, comminution of bedrock in wet-based subglacial environments promotes high rates of physical and chemical weathering because it results in a greater amount of reactive mineral surface area (Anderson, 2005; Hallet et al., 1996). In terrestrial glaciated terrains where bedrock is primarily sedimentary, metamorphic, and/or granitic in composition, outwash contains dissolved silica an order of magnitude lower than total dissolved cations (Anderson, 2005; Anderson et al., 1997). In contrast, the few studies of glaciated volcanic systems have reported outwash with dissolved SiO₂ concentrations compared to dissolved total cations (Gislason et al., 1996; Hodson et al., 2000; Yde et al., 2005). However, studies of glaciated basalts, particularly in Iceland, can be greatly influenced by inputs from hydrothermal systems, making it difficult to discern which solutes or precipitates are directly due to glacial alteration. George et al. (2007) calculated that 50% of primary weathered silica in Icelandic basalt catchments should be precipitated into secondary weathering products but did not directly tie them to glacial alteration. The glaciated volcanoes of the Cascades Range, which have little to no input from active hydrothermal systems, are an ideal location to study glacial alteration of Mars-like rocks.

We hypothesize that silica dissolution and precipitation should be predominant alteration processes occurring on actively glaciated volcanic bedrock and that SiO₂ abundance in the parent lithology should control SiO₂ dissolved in glacial meltwaters. To test this hypothesis, we examine glacial weathering of volcanic terrains at four distinct lithologic sites in the Cascades Range, USA. We use water geochemistry and infrared spectroscopy of geologic samples to examine weathering at these sites. Finally, we discuss implications of the observed alteration trends for interpreting remotely measured chemistry and mineralogy on Mars.

2. Site Descriptions and Methods

2.1. Field Sites

To investigate aqueous dissolution and precipitation of silica on a range of volcanic compositions, rock, sediment, and water samples were collected in June 2015 and July 2016 from glaciated volcanics in the Cascade Volcanic Arc (Figure 1). On Mt. Adams, two water samples were collected at Gotchen Glacier (46°9'N, 121°27'W) in June 2015. On Mt. Hood, five water samples were collected at Palmer and Eliot Glaciers (45°21'N, 121°42'W) in June 2015. On North Sister, 23 water samples were collected from Collier Glacier (44°10'N, 121°47'W) in June 2015 and July 2016, and geologic samples were collected in July 2016. On Middle Sister, 13 water and geologic samples were collected from Diller and Hayden Glaciers (44°9'N, 121°46'W) in July 2016.

Mt. Hood is a composite stratovolcano made up of dacite pyroclastic flows and lavas with smaller outcrops of basalt and andesite (Wise, 1969). Mt. Hood contains the highest silicic composition bedrock in this study (62–66% SiO₂). Eliot Glacier is a 1.6-km² glacier on the northeast flank of Mt. Hood adjacent to and partly overlying dacitic block and ash flows (Crandell, 1980; Hamilton & Havig, 2017; Jackson & Fountain, 2007).

Mt. Adams is a composite stratovolcano composed of andesite, basaltic andesite, and basalt flows, with small amounts of dacite (53–65% SiO₂; Hildreth & Lanphere, 1994). Mt. Adams represents an intermediate composition between the mafic Three Sisters complex and the silicic Mt. Hood. Gotchen Glacier is on the southeast flank of Mt. Adams (Sitts et al., 2010).

The Three Sisters volcanic complex encompasses the most mafic presently glaciated mountains in the mainland United States. North Sister and Middle Sister are composite stratovolcanoes. North Sister is primarily composed of basaltic andesite with relatively small exposures of dacite and andesite (Hildreth et al., 2012; Schmidt & Grunder, 2009). Collier Glacier on North Sister overlies a combination of basaltic andesite, andesite, and dacite (54.5–64.5% SiO₂), whereas Diller and Hayden Glaciers on Middle Sister have a more mafic substrate (52.2–56.2% SiO₂; Hildreth et al., 2012; Mercer & Johnston, 2008; Schmidt & Grunder, 2011). Subglacial and proglacial bedrock exposures in the Three Sisters are dominated by unaltered mafic units (Hildreth et al., 2012; Scudder et al., 2017).

2.2. Methods

Measurements of temperature, pH, dissolved oxygen, and conductivity were made in meltwaters at the Three Sisters using a YSI Professional Plus meter and probes (YSI, Yellow Springs, OH, USA) and at Mt. Adams and Mt. Hood using a WTW 330i meter and probe (Xylem Analytics, Weilheim, Germany). Figure 1 details the sampling locations. Dissolved silica was measured in the field using a Hach DR 1900 portable spectrophotometer (Hach, Loveland, CO, USA). All water was filtered through 25-mm-diameter, 0.2 µm polyethersulfone syringe filters (Whatman, GE Healthcare Bio-Sciences, Pittsburgh, PA, USA) and stored in presoaked (18-mΩ/cm deionized water) centrifuge tubes until analysis. Major cation (Na⁺, K⁺, Ca²⁺, and Mg²⁺) and anion (SO₄²⁻, Cl⁻, Br⁻, and F⁻) concentrations of the collected water samples were measured with ion chromatography (Dionex IonPac AS11 analytical and IonPac AG11 guard columns for anions; Dionex IonPac CS12A analytical and IonPac SG11 guard columns for cations; conductivity detection) at Arizona State University (ASU), and trace elements (e.g., Al and Fe) were determined with high-resolution inductively coupled plasma mass spectrometry (iCap-Q ICP-MS; Kinetic Energy Discrimination mode) at ASU. Dissolved inorganic carbon (DIC) was determined through elemental analysis coupled to isotope-ratio mass spectrometry (OiWet oxidation total organic carbon analyzer coupled to a Thermo Delta Plus Advantage mass spectrometer; see Havig et al., 2011, for details) at ASU. DIC represents the combination of dissolved CO_2 (as H_2CO_3) and bicarbonate (HCO_3^-), the predominant forms of DIC present over the pH ranges observed (~4.0–9.0).

Visible to near infrared (VNIR; 0.35- to 2.5- μ m spectral range; 3- to 10-nm spectral sampling) reflectance spectra of rock outcrops and proglacial sediments were collected in situ at the Three Sisters using a TerraSpec contact field spectrometer (ASD, Inc., Boulder, CO, USA). Rock and sediment samples were collected concurrently for laboratory analysis. Whenever possible, samples were collected at the same location at water samples (Figures 1 and S2). Thermal infrared (TIR; 5–25 μ m) emission spectra of these samples were collected in ASU's SpecLab facility at the Mars Space Flight Facility using a Nicolet Nexus 670 spectrometer (Thermo Fisher Scientific, Waltham, MA, USA) configured to measure emitted energy (Ruff et al., 1997). In order to increase the signal-to-noise ratio, samples were heated to 80 °C. Over the course of ~3 min, 128 spectra were collected and averaged, from 400–2,000 cm⁻¹ (5–25 μ m) with 2-cm⁻¹ spectral resolution. In order to calibrate raw data to radiance, blackbodies at 70 °C and 100 °C were measured (Christensen & Harrison, 1993). Radiance values were then normalized to the Planck curve for the derived sample temperature and transformed to emissivity spectra (Ruff et al., 1997).

In order to perform microscopic analysis, rock samples were prepared using methods similar to Minitti et al. (2007). Microscopic texture and chemistry of polished cross sections were collected with a JOEL 7600F scanning electron microscope at National Aeronautics and Space Administration Johnson Space Center. Backscattered electron imaging allowed composition and texture to be determined, and



Figure 2. (a) Mean dissolved silica in water samples at each of the four field sites. (b) pH versus dissolved silica concentration at the field sites. (c) Concentration of total dissolved major cations $(Ca^{2+}, Mg^{2+}, Na^{+}, and K^{+})$ versus dissolved silica concentration for various parts of the hydrologic systems at the Three Sisters. (d) Concentration of bicarbonate versus concentration of total cations at the Three Sisters.

qualitative chemistry was measured using electron-dispersive spectroscopy. All scanning electron microscope measurements were made at current levels between 0.5 and 0.7 nA and an accelerating voltage of 15 kV with a focused beam (~1- to 2-mm spot).

3. Results

3.1. Aqueous Geochemistry

Across the four studied sites, aqueous geochemical measurements indicate that dissolved silica ranges from 0.399 μ M to 242.2 μ Me (Figures 2a and 2b). At Diller Glacier, dissolved silica concentrations range from 0.183 to 216.4 μ M, pH ranges from 5.22 to 6.82, summed dissolved major cations (Ca²⁺, Mg²⁺, Na⁺, and K⁺) range from 15.4 to 158.2 μ M, and DIC concentrations range from 14.32 to 181.43 μ mol C/L, which equates to calculated HCO₃⁻⁻ concentrations of 3.72 to 181.43 μ eq/L. At Collier Glacier, dissolved silica concentrations range from 0.74 to 242.2 μ M, pH ranges from 4.3 to 8.36, total dissolved major cations range from 10.4 to 263.3 μ M, and DIC concentrations range from 0.00 to 221.17 μ mol C/L, which equates to calculated HCO₃⁻⁻ concentrations range from 0.00 to 221.17 μ mol C/L, which equates to calculated HCO₃⁻⁻ concentrations range from 0.00 to 221.17 μ mol C/L, which equates to calculated HCO₃⁻⁻ concentrations of 0.00 to 199.05 μ eq/L. At Gotchen Glacier, dissolved silica concentration was 10.03 μ M, and pH was measured as 5.56. One sample was collected from Gotchen Glacier from the proglacial lake (no other proglacial outwash present); it is considered representative of average proglacial runoff. At Eliot Glacier, dissolved silica concentrations range from 0.63 to 17.28 μ M, and pH measurements range from 5.57 to 5.66. For all sites, dissolved SO₄²⁻ ranges from not detectable to 1.07 μ M. Dissolved chemical concentrations are reported in Data Set S1 in the supporting information.

Aqueous silica is found in meltwater across all measured pH values (Figure 2b), but dissolved silica concentrations generally increase with increasing pH, which is consistent with silica solubility increasing with pH (e.g., McLennan, 2003). However, the effects of pH cannot explain differences between the sites. Notably, the glacier with the most mafic substrate, Diller, also exhibits the highest silica concentrations. Mean



Figure 3. (a) Context image of rock coatings on recently deglaciated mafic bedrock. (b) An scanning electron microscopederived false color image of the cross section of a rock coating. Image is composed of electron-dispersive spectroscopy chemical maps superimposed on a backscattered electron image. Colors represent elements: Fe is red; Mg is green; Si is blue; Ca is cyan; Ti is magenta; and P is yellow. (c) Representative visible to near infrared field spectrum of a rock coating, which displays similar absorptions to hydrated silica (U.S. Geological Survey Spectroscopy Laboratory Opal TM8896). (d) Representative thermal infrared spectrum of a rock coating which most resembles a combination of aluminosilicate gel and opaline silica (Michalski et al., 2003; Rampe et al., 2012).

dissolved silica concentrations increase as the underlying bedrock becomes more mafic (Figure 2a). One possible source of error with this result is the low sampling frequency at Eliot and Gotchen Glaciers, but these water samples can be considered representative of their watersheds.

Across the hydrologic systems at Three Sisters, dissolved silica concentrations are comparable to the sum concentrations of major dissolved cations (Figure 2c). Lakes, moraine-sourced springs, and proglacial streams exhibit dissolved silica concentrations that are greater than those observed in melted snow and glacial ice. The highest silica concentrations at the Three Sisters were measured in moraine-sourced spring waters.

3.2. Infrared Spectroscopy

At North and Middle Sisters, white and brown striated rock coatings on recently deglaciated lava flows (Figure 3a) were found, indicating subglacial precipitation of dissolved phases. While these types of coatings were not collected from Mt. Hood or Mt. Adams, observations by Hallet (1975) suggest that they exist throughout the Cascades. The rock coatings are characterized by pronounced lineations parallel to the

local glacial striations, suggesting that they formed in direct contact with actively sliding ice. The thickest coatings are found on the lee sides of local topographic highs on the abraded rocks, consistent with subglacial regelation films described by Hallet (1975, 1976). VNIR and TIR spectra of the coatings are both consistent with hydrated silica (Figures 3c and 3d). In the VNIR spectrum (Figure 3c), the Si-OH combination bands near 2.21 and 2.25 μ m are consistent with hydrated silica (e.g., Rice et al., 2013). In the TIR spectrum (Figure 3d), the sharp, V-shaped absorption in the Si(Al)-O stretching region (~950–1,250 cm⁻¹) is consistent with highly polymerized silicate phases (i.e., poorly crystalline secondary silicates) similar to opal or aluminosilicate gel (Michalski et al., 2006; Rampe et al., 2012). Additionally, spectra of proglacial sediments, glacial flour generated through subglacial grinding, are also consistent with a hydrated silica component (Figure S2).

3.3. Scanning Electron Microscopy

Backscattered electron images indicate that the coatings range from 30 to 225 μ m in thickness and drape over the substrate microtopography (Figures 3b and S1). Electron-dispersive spectroscopy mapping indicates that the rock coatings are enriched in silica compared to the underlying bedrock. The coatings contain internal laminations and unconformities (Figure S1). These measurements are consistent with those described by Hallet (1975) of silica-rich subglacial regelation films.

4. Discussion

Silica dissolution and precipitation is a predominant alteration process in the volcanic field sites studied here. On average, higher concentrations of dissolved silica are found at the more mafic glaciated volcanic sites compared to more felsic glaciated volcanic sites (Figure 2a). These results are most likely broadly applicable to glaciated volcanic regions, as Cascadia represents a wide range of typical volcanic compositions (Hildreth et al., 2012).

Weathering at these volcanic sites is thought to be due to one or more of the following hydrolysis (McLennan, 2003) and oxidation reactions expected for minerals found in basaltic rocks:

$$Mg_{2}SiO_{4(forsterite)} + 4H^{+} \Leftrightarrow 2 Mg^{2+} + H_{4}SiO_{4(aq)}$$
(1)

$$CaMgSi_2O_{6(diopside)} + 4H^+ + 2H_2O \Leftrightarrow Mg^{2+} + Ca^{2+} + 2H_4SiO_{4(aq)}$$
(2)

$$2NaAlSi_{3}O_{8(albite)} + 2H^{+} + 9H_{2}O \Leftrightarrow Al_{2}Si_{2}O_{5}(OH)_{4(weathered feldspar)} + 2Na^{+} + 4H_{4}SiO_{4(aq)}$$
(3)

$$2Fe_2SiO_{4(fayalite)} + O_2 + 10H_2O \Leftrightarrow 4Fe(OH)_3 + 2H_4SiO_{4(aq)}$$
(4)

Free hydrogen ions in reactions (1)–(3) are produced from dissolution of CO_2 into water and subsequent deprotonation of the resulting carbonic acid ($H_2CO_3 = > H^+ + HCO_3^-$). Reaction (4) is most likely primarily driven by subglacial microbial communities dependent on oxidation reactions for energy (e.g., Boyd et al., 2014; Hamilton et al., 2013; Mitchell et al., 2013; Raiswell et al., 2008), which are likely to be occurring at the Three Sisters based on water chemistry (Fe concentrations are comparable to major dissolved ions at some locations; see Data Set S1).

Figure 2d shows concentrations of bicarbonate with respect to concentrations of total cations at the Three Sisters. The measurements plot close to the 1:1 line, indicating that the total positive charge from cation release is balanced by negative charge produced from dissolution of CO_2 into water and subsequent deprotonation of the resulting carbonic acid ($H_2CO_3 = > H^+ + HCO_3^-$). Dissolved SO_4^{2-} is a minor component compared to bicarbonate in all sampled waters. This correlation thus implies that the majority of the weathering on glaciated volcanic bedrock at the Three Sisters is controlled by carbonic acid dissolution, as opposed to sulfide oxidation or carbonate dissolution. The full weathering reaction can be described by combining the dissolution reaction of (for example) diopside (reaction (2)) with the deprotonation of carbonic acid:

$$\mathsf{CaMgSi}_2\mathsf{O}_{6(\mathsf{diopside})} + 4\mathsf{H}_2\mathsf{CO}_3 + 2\mathsf{H}_2\mathsf{O} \Leftrightarrow \mathsf{Mg}^{2+} + \mathsf{Ca}^{2+} + 4\mathsf{HCO}_3^- + 2\mathsf{H}_4\mathsf{SiO}_{4(\mathsf{aq})} \tag{5}$$

Silica is the most abundant dissolved constituent in these glacial meltwaters, and concentrations are comparable to the total dissolved cation concentrations (Figure 2c). This is a direct result of these hydrolysis reactions (e.g., reaction (5)), where dissolved silica ($H_4SiO_{4(aq)}$) is produced in high quantities relative to other cations. In

previous studies on glaciated sedimentary or metamorphic lithologies, silica fluxes tend to be an order of magnitude lower than total cation fluxes (e.g., Anderson, 2007; Anderson et al., 1997; Hodson et al., 2002, 2000). Based on our results and previous studies, silica dissolution is the predominant reaction in these volcanic systems because of the absence of carbonate mineral dissolution, which would else result in CO₂ saturating the aqueous system and inhibiting silicate weathering (Taylor et al., 1999; Yde et al., 2005). Mafic volcanic rocks are exceptionally susceptible to silica mobilization during near-surface alteration, due to the high concentration of soluble minerals (e.g., plagioclase) versus the high concentration of relatively insoluble quartz found in felsic rocks (McLennan, 2003). Additionally, the bedrock at these sites contains reactive volcanic glass, which, Gíslason and Eugster (1987) and Hausrath et al. (2008) suggested, is more rapidly dissolved than olivine under conditions of low-temperature alteration.

Mafic rocks have a high abundance of soluble silicate phases compared to felsic rocks (McLennan, 2003). As a result, silica is expected to be highly mobile during near-surface alteration of Mars-like basaltic rocks (Soderblum, 1992), at a wide range of temperatures, pressures, and fluid chemistries. McLennan (2003) found that silica loss in basalt weathering profiles outstrip those of granodiorite and that silica is highly mobile during near-surface alteration of mafic rocks. Taylor et al. (1999) observed that terrestrial basalt weathered twice as fast as colocated felsic and intermediate bedrock, and Hodson et al. (2000) found that weathering of glacierized lcelandic basalts consumed more CO₂ than that of glacierized plutonic or metamorphic lithologies. Additionally, Stefánsson and Gíslason (2001) found that silica fluxes increased with increasing content of glass in deglaciated lcelandic basalts. This field study is consistent with previous work and better constrains the effects of residence time and cold temperatures across four distinct volcanic lithologies: With similar weathering mechanisms at work, silica is more highly mobile in the mafic, cold, Mars-like system than at the more felsic sites.

There is considerable variability in dissolved silica concentrations throughout the proglacial environment; in particular, we measured greater silica concentrations in moraine-sourced springs (Figure 2c) than in proglacial fluvial and lake settings. Moraines are composed of poorly sorted sediments with a large amount of very fine particles (glacial flour), which decreases porosity and permeability and can increase the residence time of meltwater within moraines (Hooke & Iverson, 1995; Parriaux & Nicoud, 1990). Longer residence times and the increased surface area of the fine-grained component, including volcanic glass, could contribute to the relatively high dissolved concentrations of silica compared to total dissolved cations. While a significant portion of weathering is likely happening under the glacier where meltwater interacts with basal sediments, additional alteration is occurring in the proglacial plain, where dissolved ion concentrations are greater in streams, lakes, and springs than in subglacial or supraglacial meltwaters (Figures 2c and 2d).

Our observation of striated rock glazes suggests that significant silica is being precipitated in the subglacial environment as a result of the abundant dissolved silica. The observed coatings are interpreted to be hydrated silica films formed at the ice-rock contact through subglacial pressure melting and refreezing processes known as regelation (Hallet, 1975, 1976). The coatings measured in this study contain laminations and unconformities (Figure 3b), indicating that they were most likely emplaced during successive periods of deposition. These observations support the hypothesis that silica fluxes in glacier outwash waters are lowered by subglacial precipitation of silica-rich secondary phases (Crompton et al., 2015). Supporting these chemical observations, remote sensing of the most mafic glacial catchments also suggests enhanced silica deposits in the proglacial plains (Figure S3; Scudder et al., 2017) and mineralogical analyses have found evidence for poorly crystalline, silica-rich alteration phases in mafic glacial flour (Figure S2; Rampe et al., 2017; Smith et al., 2017). Detailed discussion of the methodology can be found in the supporting information (Gillepsie, 1992).

Our results demonstrate that terrestrial cold-climate silica dissolution and precipitation is more efficient than previously reported and can be a predominant weathering process compared to dissolution of other cations, especially in glaciated mafic volcanic systems. We hypothesize that this is due to preferential silica dissolution from mafic bedrock in the absence of trace carbonates, resulting in precipitation of rock coatings composed of opaline silica and other poorly crystalline silicate alteration phases (Figure 3). The significant remaining dissolved silica and other dissolved ions that are carried out of the glacial system could also be precipitated as amorphous silicates or more crystalline secondary phases downstream of glacial environments in fluvial-lacustrine settings. Hydrated silica and other poorly crystalline silicates thus dominate the mineralogical record of glacial weathering in these systems.

Extensive evidence for present-day and past glaciation on Mars (e.g., Forget et al., 2013; Holt et al., 2008; Levy et al., 2014; Lucchitta, 1984; Pierce & Crown, 2003; Souness et al., 2012) means that chemical alteration related to glacial activity must be considered as a possible origin for silica deposits on the surface of Mars. Evidence for secondary silica and silicates on Mars has been documented by analysis of satellite VNIR and TIR spectroscopy and rover data (e.g., Ehlmann & Edwards, 2014; Milliken et al., 2008; Mustard et al., 2008; Smith et al., 2013) as well as Mars-relevant experiments (e.g., Hausrath & Brantley, 2010). Surfaces consistent with widespread poorly crystalline, high-silica phases have been identified in the northern plains using VNIR and TIR orbital spectra, and are consistent with silica rinds on volcanic glass (Horgan & Bell, 2012; Rampe et al., 2012). High-silica surfaces were identified in Hellas Basin with TIR orbital data (Bandfield, 2008). Near Valles Marineris, opaline silica was identified in VNIR orbital spectra and interpreted as evidence of lowtemperature, acidic, regionally extensive aqueous alteration (Milliken et al., 2008). Hydrated silica identified in orbital VNIR spectra at Nili Fossae has been interpreted to be the result of weathering in a hydrothermal environment (Ehlmann et al., 2009). Additionally, a silicate amorphous phase of basaltic composition has been identified with TIR data from the Sprit rover in rock coatings at Gusev Crater (Ruff & Hamilton, 2013). While secondary silica deposits have been identified globally on Mars, their origin(s) have not yet been entirely explained. Our results suggest that low-temperature weathering under a range of pH induced by seasonal snow and ice melt could, over time, produce significant silica deposits akin to those identified on Mars. This formation mechanism has not been considered before when investigating silica deposits on Mars.

5. Conclusions

Glacial weathering of terrestrial volcanic terrains is understudied and poorly understood, and thus difficult to apply to Martian alteration models. This study directly addresses this problem through water, rock, and sediment sampling across a range of glaciated bedrock compositions. Aqueous geochemical analysis of meltwater and infrared spectroscopy of geologic samples were used to evaluate alteration processes at these sites. Silica dissolution through the carbonation of mafic minerals was found to be the predominant process at each site, with greater concentrations found in more mafic terrains. This seeming incongruence can be explained by the higher solubility of olivine, feldspar, and volcanic glass compared to quartz at surface temperatures. Additionally, the greatest dissolved silica concentrations were observed in moraine-sourced spring water, with lower concentrations found in proglacial streams and lakes. We hypothesize that this is due to the higher abundance of reactive fine-grained glacial flour in moraines as well as the associated long residence times of water in these deposits. Finally, striated silica coatings were observed in glacial foregrounds, indicating subglacial silica deposition is occurring during basal sliding through the process of regelation. This suggests that significant silica cycling is occurring within the glacial system. This study underscores the need for similar investigations in other glaciated volcanic terrains to better constrain these results.

Silica dissolution and precipitation on mafic, low-carbonate rocks may be more important than previously thought on both Earth and Mars. This cycle is most likely driven by silicate-dominated weathering reactions resulting from the absence of immediately available carbonate minerals, relatively high water-rock ratios and relatively long residence times. Silica deposits on Mars should therefore not be interpreted as only hydrothermal or metamorphic in nature; rather, morphology, context, and the associated mineral assemblages should be used to rule out an origin related to glaciation, glaciofluvial alteration, or postdepositional alteration of glacial sediments. For instance, outwash plains, ancient streambeds, and lake sediments associated with past ice sheet terrains and ancient glacial moraines (Arfstrom & Hartmann, 2005; Hubbard et al., 2011) could contain alteration products, which have been transported downstream. Future landed missions and returned sample investigations should consider high-silica phases as potential signals of glacial melt in an icy climate. Detailed investigation of these phases through landed measurements and eventual sample return would greatly inform our understanding of ancient glacial alteration on the surface of Mars and could help to constrain the extent and duration of past ice sheets and glaciers and thus the nature of past climates on Mars.

References

Anderson, S. P. (2005). Glaciers show direct linkage between erosion rate and chemical weathering fluxes. *Geomorphology*, *67*, 147–157. https://doi.org/10.1016/j.geomorph.2004.07.010

Anderson, S. P. (2007). Biogeochemistry of glacial landscape systems. *Annual Review of Earth and Planetary Sciences*, *35*, 375–399. https://doi. org/10.1146/annurev.eart.35.031306.140033

Anderson, S. P., Drever, J. I., & Humphrey, N. F. (1997). Chemical weathering in glacial environments. Geology, 25(5), 399-402.

Acknowledgments

We gratefully acknowledge the Solar Systems Workings (SSW) program (Grant NNX16AG38G). We thank the National Forest Service for granting us access to the sample locations. We thank all those who assisted with fieldwork, as well as Phil Christensen, Everett Shock, and Gwyneth Gordon at ASU. We thank Marty Frisbee at Purdue University and Eve Berger at JSC. We thank the two anonymous reviewers for their helpful comments. The data used to support this paper's conclusions can be found in Data Set S1.



Bandfield, J. L. (2008). High-silica deposits of an aqueous origin in western Hellas Basin, Mars. Geophysical Research Letters, 35, L12205. https://doi.org/10.1029/2008GL033807

Boyd, E. S., Hamilton, T. L., Havig, J. R., Skidmore, M., & Shock, E. S. (2014). Chemolithotrophic primary production in a subglacial ecosystem. *Applied and Environmental Microbiology*, 80, 6146–6132. https://doi.org/10.1128/AEM.01956-14

Cassanelli, J. P., & Head, J. W. (2015). Firn densification in a Late Noachian "icy highlands" Mars: Implications for ice sheet evolution and thermal response. *Icarus*, 253, 243–255. https://doi.org/10.1016/j.icarus.2015.03.004

Christensen, P. R., & Harrison, S. T. (1993). Thermal emission spectroscopy of natural surfaces: Application to desert varnish coatings on rocks. *Journal of Geophysical Research*, *98*, 19,819–19,834. https://doi.org/10.1029/93JB00135

Colaprete, A., & Jakosky, B. M. (1998). Ice flow and rock glaciers on Mars. Journal of Geophysical Research, 103(E3), 5897–5909. https://doi.org/ 10.1029/97 JE03371

Cousins, C. R. (2015). Volcanogenic fluvial-lacustrine environments in Iceland and their utility for identifying past habitability on Mars. Life, 5, 568–586. https://doi.org/10.3390/life5010568

Cousins, C. R., Crawford, I. A., Carrivick, J. L., Gunn, M., Harris, J., Kee, T. P., et al. (2013). Glaciovolcanic hydrothermal environments in Iceland and implications for their detection on Mars. *Journal of Volcanology and Geothermal Research*, 256, 61–77. https://doi.org/10.1016/ j.jvolgeores.2013.02.009

Crandell, D. R. (1980). Recent eruptive history of Mount Hood, Oregon, and potential hazards from future eruptions, USGS Bulletin, 1492. Crompton, J. W., Flowers, G. E., Kirste, D., Hagedorn, B., & Sharp, M. J. (2015). Clay mineral precipitation and low silica in glacier meltwaters

explored through reaction-path modelling. Journal of Glaciology, 61(230), 1061–1078. https://doi.org/10.3189/2015JoG15J051 Dundas, C. M., Bramson, A. M., Ojha, L., Wray, J. J., Mellon, M. T., Byrne, S., et al. (2018). Exposed subsurface ice sheets in the Martian midlatitudes. Science, 359(6372), 199–201. https://doi.org/10.1126/science.aao1619

Ehlmann, B. L., & Edwards, C. S. (2014). Mineralogy of the Martian surface. Annual Review of Earth and Planetary Sciences, 42, 291–315. https://doi.org/10.1146/annurev-earth-060313-055024

Ehlmann, B. L., Mustard, J. F., Swayze, G. A., Clark, R. N., Bishop, J. L., Poulet, F., et al. (2009). Identification of hydrated silicate minerals on Mars using MRO-CRISM: Geologic context near Nili Fossae and implications for aqueous alteration. *Journal of Geophysical Research*, 114, E00D08. https://doi.org/10.1029/2009JE003339

Fassett, C. I., Levy, J. S., Head, J. W., & Dickson, J. L. (2014). An extended period of episodic northern and mid-latitude glaciation on Mars during the mid-to-late Amazonian: Implications for long-term obliquity history. *Geology*, 42(9), 763–766. https://doi.org/10.1130/ G35798.1

Fastook, J. L., & Head, J. W. (2014). Amazonian mid- to high-latitude glaciation on Mars: Supply-limited ice sources, ice accumulation patterns, and concentric crater fill glacial flow and ice sequestration. *Planetary and Space Science*, *91*, 60–76. https://doi.org/10.1016/ i.pss.2013.12.002

Fastook, J. L., & Head, J. W. (2015). Glaciation in the Late Noachian Icy Highlands: Ice accumulation, distribution, flow rates, basal melting, and top-down melting rates and patterns. *Planetary and Space Science, 106*, 82–98. https://doi.org/10.1016/j.pss.2014.11.028

Fastook, J. L., Head, J. W., & Marchant, D. R. (2014). Formation of lobate debris aprons on Mars: Assessment of regional ice sheet collapse and debris-cover armoring. *Icarus*, 228, 54–63. https://doi.org/10.1016/j.icarus.2013.09.025

Forget, F., Wordsworth, R., Millour, E., Madeleine, J.-B., Kerber, K., Leconte, J., et al. (2013). 3D modelling of the early Martian climate under a denser CO2 atmosphere: Temperatures and CO2 ice clouds. *Icarus*, 222, 81–99. https://doi.org/10.1016/j.icarus.2012.10.019

George, R. B., Reynolds, B. C., West, A. J., Burton, K. W., & Halliday, A. N. (2007). Silicon isotope variations accompanying basalt weathering in Iceland. *Earth and Planetary Science Letters*, 261, 476–490. https://doi.org/10.1016/j.epsl.2007.07.004

Gillepsie, A. R. (1992). Enhancement of multispectral thermal infrared images: Decorrelation contrast stretching. *Remote Sensing of Environment*, 42, 147–155.

Gíslason, S., Arnorsson, S., & Armannsson, H. (1996). Chemical weathering of basalt in southwest Iceland: Effects of runoff, age of rocks, and vegetative/glacial cover. American Journal of Science, 296, 837–907.

Gíslason, S. R., & Eugster, H. (1987). Meteoric water-basalt interaction: I. A laboratory study. *Geochimica et Cosmochimica Acta*, *51*, 2827–2840. Hallet, B. (1975). Subglacial silica deposits. *Nature*, *254*, 682–683.

Hallet, B. (1976). Deposits formed by subglacial precipitation of CaCO₃. Geological Society of America Bulletin, 87, 1003–1015.

Hallet, B., Hunter, L., & Bogen, J. (1996). Rates of erosion and sediment evacuation by glaciers: A review of field data and their implications. *Global and Planetary Change*, 12, 213–235.

Hamilton, T. L., & Havig, J. (2017). Primary productivity of snow algae communities on stratovolcanoes of the Pacific Northwest. *Geobiology*, 1–16. https://doi.org/10.1111/gbi.12219

Hamilton, T. L., Peters, J. W., Skidmore, M. L., & Boyd, E. S. (2013). Molecular evidence for an active endogenous microbiome beneath glacial ice. *The ISME Journal*, *7*, 1402–1412. https://doi.org/10.1038/ismej.2013.31

Hausrath, E. M., & Brantley, S. L. (2010). Basalt and olivine dissolution under cold, salty, and acidic conditions: What can we learn about recent aqueous weathering on Mars? *Journal of Geophysical Research*, *115*, E12001. https://doi.org/10.1029/2010JE003610

Hausrath, E. M., Treiman, A. H., Vicenzi, E., Bish, D. L., Blake, D., Sarrazin, P., et al. (2008). Short- and long-term olivine weathering in Svalbard: Implications for Mars. Astrobiology, 8(6), 1079–1092. https://doi.org/10.1089/ast.2007.0195

Havig, J. R., Meyer-Dombard, D. R., Raymond, J., & Shock, E. L. (2011). Merging isotopes and metagenomics: Coupling biofilm C and N isotopes and metagenomics in a siliceous sinter-depositing hot spring. *Journal of Geophysical Research*, 116, G01005. https://doi.org/10.1029/ 2010JG001415

Head, J. W., Marchant, D. R., Agnew, M. C., Fassett, C. I., & Kreslavsky, M. A. (2006). Extensive valley glacier deposits in the northern mid-latitudes of Mars: Evidence for late Amazonian obliquity-driven climate change. *Earth and Planetary Science Letters*, 241(3–4), 663–671. https://doi.org/10.1016/j.epsl.2005.11.016

Head, J. W., Neukum, G., Jaumann, R., Hiesinger, H., Hauber, E., Carr, M., et al. (2005). Tropical to mid-latitude snow and ice accumulation, flow and glaciation on Mars. *Nature*, 434, 346–351. https://doi.org/10.1038/nature03359

Hildreth, W., Fierstein, J., & Calvert, A.T. (2012). Geologic map of the Three Sisters volcanic cluster, Cascade Range, Oregon. USGS Science Investigations Map 3186.

Hildreth, W., & Lanphere, M. A. (1994). Potassium-argon geochronology of a basalt-andesite-dacite arc system: The Mount Adams volcanic field, Cascade Range of southern Washington. *Geological Society of America Bulletin*, 106, 1413–1429.

Hodson, A., Porter, P., Lowe, A., & Mumford, P. (2002). Chemical denudation and silicate weathering in Himalayan glacier basins: Batura Glacier, Pakistan. Journal of Hydrology, 262, 193–208. https://doi.org/10.1016/S0022-1694(02)00036-7



Hodson, A., Tranter, M., & Vatne, G. (2000). Contemporary rates of chemical denudation and atmospheric CO₂ sequestration in glacier basins: An Arctic perspective. *Earth Surface Processes and Landforms*, 25, 1447–1471. https://doi.org/10.1002/1096-9837(200012)25:13%3C1447:: AID-ESP156%3E3.0.CO;2-9

Holt, J. W., Safaeinili, A., Plaut, J. J., Head, J. W., Phillips, R. J., Seu, R., et al. (2008). Radar sounding evidence for buried glaciers in the southern mid-latitudes of Mars. *Science*, 322, 1235–1238. https://doi.org/10.1126/science.1164246

Hooke, R. L., & Iverson, N. R. (1995). Grain-size distribution in deforming subglacial tills: Role of grain fracture. *Geology*, 23(1), 57–60. Horgan, B., & Bell, J. F. (2012). Widespread weathered glass on the surface of Mars. *Geology*, 40, 391–394. https://doi.org/10.1130/G32755.1

Hubbard, B., Milliken, R. E., Kargel, J. S., Limaye, A., & Souness, C. (2011). Geomorphological characterisation and interpretation of a mid-latitude glacier-like form: Hellas Planitia, Mars. *lcarus*, 211, 330–346. https://doi.org/10.1016/j.icarus.2010.10.021

Jackson, K. M., & Fountain, A. G. (2007). Spatial and morphological change on Eliot Glacier, Mount Hood, Oregon, USA. Annals of Glaciology, 46, 222–226. https://doi.org/10.3189/172756407782871152

- Kite, E. S., Halevy, I., Kahre, M. A., Wolff, M. J., & Manga, M. (2013). Seasonal melting and the formation of sedimentary rocks on Mars, with predictions for the Gale Crater mound. *Icarus*, 223, 181–210. https://doi.org/10.1016/j.icarus/2012.11.034
- Kite, E. S., Williams, J.-P., Lucas, A., & Aharonson, O. (2014). Low palaeopressure of the Martian atmosphere estimated from the size distribution of ancient craters. *Nature Geoscience*, 7, 335–339. https://doi.org/10.1038/ngeo2137

Levy, J. S., Fassett, C. I., Head, J. W., Schwartz, C., & Watters, J. L. (2014). Sequestered glacial ice contribution to the global Martian water budget: Geometric constraints on the volume of remnant, midlatitude debris-covered glaciers. *Journal of Geophysical Research: Planets*, 119, 2188–2196. https://doi.org/10.1002/2014JE004685

Levy, J. S., Head, J. W., & Marchant, D. R. (2007). Lineated valley fill and lobate debris apron stratigraphy in Nilosyrtis Mensae, Mars: Evidence for phases of glacial modification of the dichotomy boundary. *Journal of Geophysical Research*, 112, E08004. https://doi.org/10.1029/ 2006JE002852

Levy, J. S., Head, J. W., & Marchant, D. R. (2010). Concentric crater fill in the northern mid-latitudes of Mars: Formation processes and relationships to similar landforms of glacial origin. *Icarus*, 209(2), 390–404. https://doi.org/10.1016/j.icarus.2010.03.036

Lucchitta, B. K. (1984). Ice and debris in the fretted terrain. Mars. Journal of Geophysical Research, 89, B409–B418.

Mangold, N. (2003). Geomorphic analysis of lobate debris aprons on Mars at Mars Orbiter Camera scale: Evidence for ice sublimation initiated by fractures. *Journal of Geophysical Research*, 108(E4), 8021. https://doi.org/10.1029/2002JE001885

McLennan, S. M. (2003). Sedimentary silica on Mars. Geology, 31(4), 315-318.

Mercer, C. N., & Johnston, A. D. (2008). Experimental studies of the P-T-H2O near-liquidus phase relations of basaltic andesite from North Sister Volcano, High Oregon Cascades: Constraints on lower-crustal mineral assemblages. *Contributions to Mineralogy and Petrology*, 155, 571–592. https://doi.org/10.1007/s00410-007-0259-8

Michalski, J. R., Kraft, M. D., Diedrich, T., Sharp, T. G., & Christensen, P. R. (2003). Thermal emission spectroscopy of the silica polymorphs and consideration for remote sensing of Mars. *Geophysical Research Letters*, *30*(19), 2008. https://doi.org/10.1029/2003GL018354

Michalski, J. R., Kraft, M. D., Sharp, T. G., & Christensen, P. R. (2006). Effects of chemical weathering on infrared spectra of Columbia River basalt and spectral interpretations of Martian alteration. *Earth and Planetary Science Letters*, 248, 822–829. https://doi.org/10.1016/ j.epsl.2006.06.034

Milliken, R. E., Mustard, J. F., & Goldsby, D. L. (2003). Viscous flow features on the surface of Mars: Observations from high-resolution Mars Orbiter Camera (MOC) images. Journal of Geophysical Research, 108(E6), 5057. https://doi.org/10.1029/2002JE002005

Milliken, R. E., Swayze, G. A., Arvidson, R. E., Bishop, J. L., Clark, R. N., Ehlmann, B. L., et al. (2008). Opaline silica in young deposits on Mars. *Geology*, 36, 847–850. https://doi.org/10.1130/G2497A.1

Minitti, M. E., Weitz, C. M., Lane, M. D., & Bishop, J. L. (2007). Morphology, chemistry, and spectral properties of Hawaiian rock coatings and implications for Mars. Journal of Geophysical Research, 112, E05015. https://doi.org/10.1029/2006JE002839

Mitchell, A. C., Lafrenière, M. L., Skidmore, M. L., & Boyd, E. S. (2013). Influence of bedrock mineral composition on microbial diversity in a subglacial environment. *Geology*, 41, 855–858. https://doi.org/10.1130/G34194

Mustard, J. F., Murchie, S. L., Pelkey, S. M., Ehlmann, B. L., Milliken, R. E., Grant, J. A., et al. (2008). Hydrated silicate minerals on Mars observed by the Mars Reconnaissance Orbiter CRISM instrument. *Nature*, 454, 305–309. https://doi.org/10.1038/nature07097

Parriaux, A., & Nicoud, G. F. (1990). Hydrological behavior of glacial deposits in mountainous areas. *Hydrology of Mountainous Areas*, 190, 291–312.

Pierce, T. L., & Crown, D. A. (2003). Morphologic and topographic analyses of debris aprons in the eastern Hellas region, Mars. *lcarus*, 163(1), 46–65. https://doi.org/10.1016/S0019-1035(03)00046-0

Plaut, J. J., Safaeinili, A., Holt, J. W., Phillips, R. J., Head, J. W. III, Seu, R., et al. (2009). Radar evidence for ice in lobate debris aprons in the mid-northern latitudes of Mars. *Geophysical Research Letters*, 36, L02203. https://doi.org/10.1029/2008GL036379

Raiswell, R., Benning, L. G., Davidson, L., & Tranter, M. (2008). Nanoparticulate bioavailable iron minerals in icebergs and glaciers. *Mineralogical Magazine*, 72, 345–348. https://doi.org/10.1180/minmag.2008.072.1.345

Rampe, E. B., Horgan, B., Scudder, N. A., Smith, R. J., & Rutledge, A. M. (2017). Mineralogy of rock flour in glaciated volcanic terrains: An analog for a cold and icy early Mars. Paper presented at 48th Lunar and Planetary Science Conference, Lunar and Planetary Institute, The Woodlands, TX.

Rampe, E. B., Kraft, M. D., Sharp, T. G., Golden, D. C., Ming, D. W., & Christensen, P. R. (2012). Allophane detection on Mars with Thermal Emission Spectrometer data and implications for regional-scale chemical weathering processes. *Geology*, 40, 995–998. https://doi.org/ 10.1130/G33215.1

Rice, M. S., Cloutis, E. A., Bell, J. F., Bish, D. L., Horgan, B. H., Mertzman, S. A., et al. (2013). Reflectance spectra diversity of silica-rich materials: Sensitivity to environment and implications for detections on Mars. *lcarus*, 223(1), 499–533. https://doi.org/10.1016/ j.icarus.2012.09.021

Rossbacher, L. A., & Judson, S. (1981). Ground ice on Mars: Inventory, distribution, and resulting landforms. *Icarus*, 45(1), 39–59.

Ruff, S. W., Christensen, P. R., Barbera, P. W., & Anderson, D. L. (1997). Quantitative thermal emission spectroscopy of minerals: A laboratory technique for measurement and calibration. *Journal of Geophysical Research*, 102(B7), 14,899–14,913. https://doi.org/10.1029/ 97JB00593

- Ruff, S. W., & Hamilton, V. E. (2013). Amorphous Mars: Interpreting growing evidence for poorly/non-crystalline phases in Martian materials. Paper presented at 44th Lunar and Planetary Science Conference, Lunar and Planetary Institute, The Woodlands, TX.
- Schmidt, M. E., & Grunder, A. L. (2009). The evolution of North Sister: A volcano shaped by extension and ice in the central Oregon Cascade Arc. *Geological Society of America Bulletin*, 121(5/6), 643–662. https://doi.org/10.1130/B26442.1

Schmidt, M. E., & Grunder, A. L. (2011). Deep mafic roots to arc volcanoes: Mafic recharge and differentiation of basaltic andesite at North Sister Volcano, Oregon Cascades. *Journal of Petrology*, *52*, 603–641. https://doi.org/10.1093/petrology/egq094



Scudder, N. A., Horgan, B., Rutledge, A. M., & Rampe, E. B. (2017). Using composition to trace glacial, fluvial, and aeolian sediment transport in a Mars-analog glaciated volcanic system. Paper presented at 48th Lunar and Planetary Science Conference, Lunar and Planetary Institute, The Woodlands, TX.

Sharp, R. P. (1973). Mars: Fretted and chaotic terrains. Journal of Geophysical Research, 78(20), 4073–4083. https://doi.org/10.1029/ JB078i020p04073

Sitts, D. J., Fountain, A. G., & Hoffman, M. J. (2010). Twentieth century glacier change on Mount Adams, Washington, USA. Northwest Science (Northwest Scientific Association), 84, 378–385. https://doi.org/10.3955/046.084.0407

Smith, M. R., Bandfield, J. L., Cloutis, E. A., & Rice, M. S. (2013). Hydrated silica on Mars: Combined analysis with near-infrared and thermal-infrared spectroscopy. *Icarus*, 223(2), 633–648. https://doi.org/10.1016/j.icarus.2013.01.024

Smith, R. J., Horgan, B., Rampe, E., Dehouck, E., & Morris, R. V. (2017). X-ray amorphous phases in terrestrial analog volcanic sediments: Implications for amorphous phases in Gale Crater, Mars. Paper presented at 48th Lunar and Planetary Science Conference, Lunar and Planetary Institute, The Woodlands, TX.

Soderblum, L. A. (1992). The composition and mineralogy of the Martian surface from spectroscopic observations: 0.3 mm to 50 mm. In H. H. Kieffer, B. M. Jakosky, C. W. Snyder, & M. S. Matthews (Eds.), *Mars* (pp. 557–593). Tucson: University of Arizona Press.

Souness, C., Hubbard, B., Milliken, R. E., & Quincey, D. (2012). An inventory and population-scale analysis of Martian glacier-like forms. *Icarus*, 217, 243–255. https://doi.org/10.1016/j.icarus.2011.10.020

Squyres, S. W. (1978). Martian fretted terrain: Flow of erosional debris. Icarus, 34(3), 600-613.

Squyres, S. W. (1979). The distribution of lobate debris aprons on Mars. *Journal of Geophysical Research*, 84, 8087–8096. https://doi.org/ 10.1029/JB084iB14p08087

Squyres, S. W., & Carr, M. H. (1986). Geomorphic evidence for the distribution of ground ice on Mars. Science, 231(4735), 249-252.

Stefánsson, A., & Gíslason, S. R. (2001). Chemical weathering of basalts, southwest Iceland: Effect of rock crystallinity and secondary minerals on chemical fluxes to the ocean. *American Journal of Science*, 301, 513–556.

Taylor, A. S., Lasaga, A. C., & Blum, J. D. (1999). Effect of lithology on silicate weathering rates. In H. Ármannsson (Ed.), *Geochemistry of the Earth's surface* (pp. 127–128). Rotterdam: Balkema.

Wise, W. S. (1969). Geology and petrology of the Mt. Hood area: A study of high Cascade volcanism. *Geological Society of America Bulletin*, 80, 969–1006.

Wordsworth, R., Forget, F., Millour, E., Head, J. W., Madeleine, J.-B., & Charnay, B. (2013). Global modelling of the early Martian climate under a denser CO₂ atmosphere: Water cycle and ice evolution. *Icarus*, 222, 1–19. https://doi.org/10.1016/j.icarus.2012.09.036

Yde, J. C., Knudsen, N. T., & Nielsen, O. B. (2005). Glacier hydrochemistry, solute provenance, and chemical denudation at a surge-type glacier in Kuannersuit Kuussuat, Disko Island, West Greenland. *Journal of Hydrology*, 300, 172–187. https://doi.org/10.1016/j.hydrol.2004.06.008